Author’s response to reviews

Title: Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome

Authors:

Deepak Goyal (docdeepakgoyal@yahoo.com)
Dinesh K Yadav (dineshkumar169@yahoo.co.in)
Umesh Shukla (drumeshshukla@gmail.com)
Sidharth K Sethi (sidsdoc@gmail.com)

Version: 4 Date: 8 July 2010

Author’s response to reviews: see over
To,
The Editor
J Medical Case Reports

Sub: Submission of the revisions to the “Case report” titled “Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome”.

Respected Sir/Madam,

We are sending the revised case report titled “Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome”, along with point by point response to the concerns. All the changes in the manuscript have been highlighted in red.

We look forward to its acceptance for publication in the esteemed journal.

Thanking you

Yours Sincerely

Dr Deepak Goyal
Senior Resident
PGIMER & associated Dr. RML Hospital
New Delhi, India
E mail: docdeepakgoyal@yahoo.com
Editorial comments
We thank you for your kind comments.

Please place the authors contribution section, competing interests and consent information on the last page of the manuscript.

The authors contribution section, competing interests and consent information have been placed on the last page of manuscript.

Contributions: DG, SKS and US were involved in the care of the patients, relevant investigations and prepared the manuscript. DKY supervised and helped in revision of the manuscript. All authors read and approved the final manuscript.

Funding: None

Competing interests: None stated

Consent: An informed consent for the case report and pictures was taken from the parents.

Your abstract should be structured into the following three sections:

Introduction An introduction about why this case is important and needs to be reported. Please include information on whether this is the first report of this kind in the literature.

Case presentation Brief details of what the patient presented with, including the patient’s age, sex and ethnic background.

Conclusion A brief conclusion of what the reader should learn from the case report and what the clinical impact will be. Is it an original case report of interest to a particular clinical speciality of medicine or will it have a broader clinical impact across medicine? Please include information on how it will significantly advance our knowledge of a particular disease etiology or drug mechanism.

Abstract:

Introduction: We report an unusual association of Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome. This association has never been reported in medical literature.

Case presentation: A 9 year old, Indian girl, was referred for growth retardation, mental retardation, lax joints, generalized hypertrichosis, hypoplastic fifth fingernail and toe nails. A thorough medical
examination and evaluation showed the phenotypic features of Coffin Siris Syndrome, with Mayer-Rokitansky-Kuster-Hauser syndrome on radiological evaluation. The karyotype of the child was normal.

**Conclusion:**
In an unexplained mental retardation with facies suggestive of Coffin Siris syndrome, this association with Mayer-Rokitansky-Küster-Hauser syndrome should be looked for, and the patient should be evaluated for the same. Both of these syndromes may have a common pathogenesis, yet unknown. This case report has broad implications as similar cases in future may give insights into the pathogenesis of both these syndromes.

*We have made sure that the manuscript follows the journal style.*